Author's response to reviews

Title: A cost minimisation analysis of a telepaediatric otolaryngology service

Authors:

Cathy Q Xu (c.xu@uq.edu.au)
Anthony C Smith (a.smith@pobox.com)
Paul A Scuffham (p.scuffham@griffith.edu.au)
Richard Wootton (r.wootton@pobox.com)

Version: 5 Date: 28 January 2008

Author's response to reviews:

25 January 2008

To the Editor

BMC - Health Services Research

Dear Sir/Madam,

Re: Formatting changes and minor revisions - MS: 5266354814714116

A cost minimisation analysis of a telepaediatric otolaryngology service

Thank you for the opportunity to review the final proof of our paper. We have made formatting changes as requested. We also made some minor changes to improve the readability of the paper. This includes revision of the costs which included a document camera for viewing hard copies of documents in the telemedicine service. These updates have not changed the results and conclusion of our study. We look forward to seeing the published article in the BMC journal.

Please find attached the revised manuscript with the specified amendments.

Regards,

Dr Cathy Xu